Author’s response to reviews

Title: CARDIOPULMONARY RESUSCITATION COMPLICATIONS ENCOUNTERED IN FORENSIC AUTOPSY CASES

Authors:

Aspasia Deliliga (aspa_del@hotmai.com)
Fotios Chatzinikolaou (fchatzin@yahoo.gr)
Dimitrios Koutsoukis (dimitrios.k.koutsoukis@gmail.com)
Ioannis Chrysovergis (j.chrysovergis@gmail.com)
Polychronis Voultsos (pvoultsos@auth.gr)

Version: 1 Date: 12 Feb 2019

Author’s response to reviews:

5 Feb 2019

Dear Editor,

I re-submit our revised paper entitled: CARDIOPULMONARY RESUSCITATION COMPLICATIONS ENCOUNTERED IN FORENSIC AUTOPSY CASES (Manuscript number: EMMD-D-18-00147).

On behalf of all the authors of the manuscript I am very grateful to the two reviewers for their careful and meticulous reading of the paper. We kindly acknowledge the reviewers’ comments and suggestions very much, which are valuable in improving the quality of our manuscript. We have tried my best to address the comments raised by the reviewers and to revise the manuscript in accordance with the precious and helpful major comments they raised.

Many changes have been made in different areas of the manuscript. Note, however, that the major changes in the revised manuscript are highlighted in yellow color.
Moreover, we carried out a scientific (and language) editing of our paper, hoping that it has helped to improve the manuscript. Therefore, we took a deeper look at the content of my paper focusing on the parts that are essential to producing an improved paper, namely, to making it much easier to understand, better organized and more suitable for the audience (readership). Finally, a careful proof-reading of the manuscript was carried out to minimize typographical, grammatical, and bibliographical errors.

Dear Editor, thank you for your consideration of our manuscript and the opportunity you have given us to revise and resubmit our paper. I hope that we have increased the chance of getting our manuscript published. However, the final decision is yours to make.

I look forward to hearing from you.

Sincerely yours

Dr Polychronis Voultosos
Assistant Professor of Medical Ethics
Aristotle University of Thessaloniki (Faculty of Medicine)
Greece
(Corresponding Author)